%% Boxes_and_Glue.tex
%
%% Copyright 2010 LAFKON/Christoph Haag 

\documentclass[8pt]{scrbook}
\usepackage{i/a/utils/latex/sty/LSI-191x235+crop}
\usepackage{i/a/workshop/bag/Boxes_and_Glue}

\newcommand{\fpath}{i/a/workshop/bag}
\makeindex

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}

%\titlepages % NEWCOMMAND DEFINED IN Boxes_and_Glue.sty
%\input{documentation.tex}



\input{test.tex}



%\printindex
%\nocite{sokal:1996:transgressing}
%\nocite{genter:1996:tami}
%\nocite{debruijn:2006:dd}
%\nocite{raskin:1993:nogui}
%\nocite{osp:2006:dtfd}
%\bibliographystyle{i/a/utils/latex/sty/bst/unsrtmod}
%\bibliography{i/a/utils/latex/references}

\end{document}
